\begin{table}%t2
%%\centering
\par
\caption{Infrared excess in V838~Mon: data and results.
See text for explanations of the columns.}
\label{ir_t}
%\centerline
{
\begin{tabular}{@{} p{0.4cm} p{2.15cm} p{0.6cm} p{0.6cm} p{0.6cm} p{0.8cm} p{0.7cm} @{}}
\hline\hline
  Day & Data & $T_{\rm BB,s}$ & $T_{\rm BB,d}$ & $-$lg~$\theta_{\rm d}$ & 
$R_{\rm d}/R_{\odot}$ & $L_{\rm d}/L_{\rm s}$ \\
\hline
   ~15. & ${\it BVRI}$(clk3) & 5200. \\
        & ${\it LMN}$(lrr4) \\
   28.5 & ${\it BVRI}$(kls2) & 4650. \\
        & ${\it JHK}$(lrr4) \\
        & 9--12$\mu$m(klk2) \\
   ~42. & $U$--$L$(mhk2) & 6880. \\
        & 10--12$\mu$m(klk2) \\
  258. & $B$--$L$(clm5) & 1840. \\
        & 8--18$\mu$m(wc2) \\
  395. & $B$--$L$(clm5)  & 2220. &  800. & ~~7.75 & ~~6.3E3 & 0.29 \\
        & $J$--$N$(lrr4) \\
  650. & $B$--$L$(clm5)  & 2520. &  600. & ~~7.43 & 13.1E3 & 0.54 \\
        & ${\it LMN}$(lrp3) \\
        & 9--19$\mu$m(tp3) \\
\hline
\end{tabular}}
\medskip
Data references: 
clk3~--~Crause et~al. (\cite{lisa03});
clm5~--~Crause et~al. (\cite{lisa04});
klk2~--~Kaeufl et~al. (\cite{kaufl});
kls2~--~Kimeswenger et~al. (\cite{kimes});
lrp3~--~Lynch et~al. (\cite{lynch03});
lrr4~--~Lynch et~al. (\cite{lynch04});
mhk2~--~Munari et~al. (\cite{munari});
tp3~--~Tapia \& Persi (\cite{tapia});
wc2~--~Watson \& Costero (\cite{watcos}).
\end{table}